Author's response to reviews

Title: Surgical and conservative treatment of patients with congenital scoliosis: A search for long-term results

Authors:

Angelos Kaspiris (angkaspiris@hotmail.com)
Theodoros B Grivas (grivastb@vodafone.net.gr)
Hans-Rudolf Weiss (hr.weiss@skoliose-dr-weiss.com)
Deborah Turnbull (Deborahturnbull@nhs.net)

Version: 10 Date: 4 May 2011

Author's response to reviews: see over
To the Editor

Dear Editor,

I would like to thank you about our co-operation during the submission process. We are submitting a new version of our manuscript, based on the report of the last reviewer, directly to you.

The corrections of our manuscript are as follows:
1. Page 9: Unfortunately, the phrase “In addition” is written not only twice but three times. It has been replaced by “Furthermore” in the second sentence and by “Finally” in the third.
2. Figure 3: the word “curve” has been added in the text after all the measurements.
3. Figure 4: the age of the patient the final follow-up, during which the image’s clinical signs were observed, was clarified.

I am looking forward to hearing from you soon.

Kind regards
Dr Angelos Kaspiris MD, MPhil